
   

 

   
 

EATM ACTION ITEMS & Follow Ups – January 16th 2023 

 

 
 
 
 
  

Nr Item Responsible Comment Status 

1 Procedure for galleries access 
B. Rae, J. 

Bernhard  

New version under approval and to be merged with 

EDMS 1748425. EDMS 2391345 
Ongoing 

2 
BE-DSO recommendation for 

audible alarm in GIF++ 

M. Jaekel, N. 

Charitonidis 

Meeting done and recommendation to add some 

alarm units on the GIF++ radiation monitors. Cost 

and installation scheduled being checked. 

Ongoing 

3 
Vacuum implementation in 

F61.MBXHD025 

P. Boisseaux-

Bourgeios, D. 

Banerjee 

Vacuum to be installed in the F61.MBXHD025 for 

CHIMERA/HEARTS to reduce material. Integration 

of the beam instrumentation with beamline vacuum 

also to be checked.  

Ongoing 

https://edms.cern.ch/document/SPSX-S-OP-0001/0.3/approvalAndComments


   

 

   
 

NA-CONS Related Activities 
 

Nr Item Responsible Comment Status 

1 
ATEX ventilation 

works 
EN-CV 

Reception of cubicles from Spain in wk10, installation 

works from wk12-wk15 and commissioning in wk16 

(Ap'24) , No impact during RUN period. 

Ongoing 

2 

Consolidation of the gas 

mixing area on the Jura 

side 

BE-EA Activity postponed to YETS 2024/25. Postponed 

3 

Consolidation of the 

chilled water 

piping/distribution in 

EHN1 

EN-CV 

Advancing well, more resources allocated, completion 

foreseen 22.03.24 (wk12). WSS for NA-CONS 

confirmed the feedback. Close monitoring is being 

done. 

2023/24 

YETS 

4 

EHN2 Underground 

Galleries Ventilation 

Installation    

EN-CV  

Mechanical activities are finished. Commissioning 

expected wk11. The lights will have to be installed 

during a Wednesday in the future because supply 

issues. EN-CV will do the commissioning without 

them. 

Ongoing 

5 
Installation of the fire 

doors in GHN21/22 
EN-ACE 

Works on-going. Installation of fire doors done 

without insulation around doors in EHN1-BA81 

accumulating delay of 2 weeks. Complete 

installation of fire doors EHN2/GHN2 by end 

of this week (wk10). EN-AA to communicate 

with EN-CV for interlocking of the systems. 

Ongoing 

6 
Installation/test of new 

rad-hard monitor 
BE-EA / SY-BI  

All necessary works have been done. Cabling 

completed, installation on-going and foreseen 

completion wk10. 

Ongoing 



   

 

   
 

 

7 

Consolidation of the fire 

sprinkler station in 

BA80/PA80/TA802 

EN-AA 

New sprinkler station and PA80 installation 

completed, site commissioning test foreseen in 

wk10, Both systems will be operational wk11.  

Works for TA801, PGT802 & GL802 

postponed to EYETS24-25. 

Ongoing 

8 
Fire detection 

underground 
EN-AA 

 Works completed for EYETS23-24. 

Installation DI GL812, Installation DI GL811 

(GL801 - GHN101) : 100 % completion for 

electrical works . Piping works postponed to 

(E)YETS 24-25. 

Ongoing 

9 
Alignment of H6/H8 

beamlines 
BE-GM Activity is completed Done 


